# UC San Diego UC San Diego Previously Published Works

## Title

Identifying contributors to disparities in patient access of online medical records: examining the role of clinician encouragement.

## Permalink

https://escholarship.org/uc/item/8bm7q7bk

**Journal** JAMIA Open, 6(3)

## Authors

Sisk, Bryan Lin, Sunny Balls-Berry, Joyce <u>et al.</u>

Publication Date 2023-10-01

## DOI

10.1093/jamiaopen/ooad049

Peer reviewed

# **Research and Applications**

# Identifying contributors to disparities in patient access of online medical records: examining the role of clinician encouragement

# Bryan A. Sisk (1<sup>,2,\*</sup>, Sunny Lin (2<sup>,</sup>, Joyce (Joy) E. Balls-Berry<sup>3</sup>, Argentina E. Servin<sup>4</sup>, and Jennifer W. Mack<sup>5,6</sup>

<sup>1</sup>Division of Hematology/Oncology, Department of Pediatrics, Washington University School of Medicine, St. Louis, Missouri, USA <sup>2</sup>General Medical Sciences, Department of Medicine, Washington University School of Medicine, St. Louis, Missouri, USA

<sup>3</sup>Department of Neurology, Washington University School of Medicine, St. Louis, Missouri, USA

<sup>4</sup>Department of Medicine, University of California San Diego, San Diego, California, USA

<sup>5</sup>Department of Pediatric Oncology and Division of Population Sciences, Dana-Farber Cancer Institute, Boston, Massachusetts, USA <sup>6</sup>Division of Pediatric Hematology/Oncology, Boston Children's Hospital, Boston, Massachusetts, USA

\*Corresponding Author: Bryan A. Sisk, MD, Division of Hematology/Oncology, Department of Pediatrics, Washington University School of Medicine, 4523 Clayton Avenue, 8005, St. Louis, MO 63110, USA; siskb@wustl.edu

Bryan A. Sisk and Sunny Lin are co-first authors.

#### ABSTRACT

**Objective:** The aim of this study was to understand the influence of clinician encouragement and sociodemographic factors on whether patients access online electronic medical records (EMR).

**Materials and Methods:** We analyzed 3279 responses from the Health Information National Trends Survey 5 cycle 4 survey, a cross-sectional, nationally representative survey administered by the National Cancer Institute. Frequencies and weighted proportions were calculated to compare clinical encouragement and access to their online EMR. Using multivariate logistic regression, we identified factors associated with online EMR use and clinician encouragement.

**Results:** In 2020, an estimated 42% of US adults accessed their online EMR and 51% were encouraged by clinicians to access their online EMR. In multivariate regression, respondents who accessed EMR were more likely to have received clinician encouragement (odds ratio [OR], 10.3; 95% confidence interval [CI], 7.7–14.0), college education or higher (OR, 1.9; 95% CI, 1.4–2.7), history of cancer (OR, 1.5; 95% CI, 1.0–2.3), and history of chronic disease (OR, 2.3; 95% CI, 1.7–3.2). Male and Hispanic respondents were less likely to have accessed EMR than female and non-Hispanic White respondents (OR, 0.6; 95% CI, 0.5–0.8, and OR, 0.5; 95% CI, 0.3–0.8, respectively). Respondents receiving encouragement from clinicians were more likely to be female (OR, 1.7; 95% CI, 1.3–2.3), have college education (OR, 1.5; 95% CI, 1.1–2.0), history of cancer (OR, 1.8; 95% CI, 1.3–2.5), and greater income levels (OR, 1.8–3.6).

**Discussion:** Clinician encouragement of patient EMR use is strongly associated with patients accessing EMR, and there are disparities in who receives clinician encouragement related to education, income, sex, and ethnicity.

Conclusions: Clinicians have an important role to ensure that all patients benefit from online EMR use.

#### LAY SUMMARY

Online medical records are a widely available tool that might help to improve communication and help patients to better take care of their health. But past studies have shown that certain types of patients are less likely to use their online records, especially patients from racial and ethnic minorities and patients with lower income or less education. None of these studies tried to figure out the clinicians' role in encouraging patients to use these online medical records. In this study, we wanted to understand whether clinicians encouraging patients to use online medical records seemed to influence whether patients actually used these online records. We analyzed 3279 responses from a national survey that was administered across the United States by the National Cancer Institute. We found that participants who accessed their online medical records were also more likely to have received encouragement from their clinician. Participants who received encouragement from their clinicians were more likely to be female, have college education, a history of cancer, and higher-income levels. Taken together, our results suggest that clinicians have an important role to ensure that all patients benefit from online medical record access.

Key words: Electronic medical record, online patient portals, communication, physician-patient relationship, healthcare disparities

#### **BACKGROUND AND SIGNIFICANCE**

Patient portals represent a widely available tool that might address communication barriers and improve healthcare delivery. Since April 2021, the final rule for implementation of the 21st Century Cures Act prevents healthcare systems from information blocking, defined as any practice or policy that "interferes with access, exchange, or use of the electronic health information from patients and their legal guardians."<sup>1</sup>

Received: 17 October 2022. Revised: 30 April 2023. Editorial Decision: 26 June 2023. Accepted: 28 June 2023

© The Author(s) 2023. Published by Oxford University Press on behalf of the American Medical Informatics Association.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/ licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com As a result, healthcare institutions across the United States provide access to electronic health information (EHI) through online patient portals. Given the wide availability of portals and the federal mandate for transparency, portals represent a powerful tool to support communication and care delivery in the US healthcare system.

Patients who access their EHI through portals report several benefits. In adult cancer, patients report that using portals and reading notes helps them to make sense of their diagnosis and treatment, maintain communication with oncology clinicians, and engage with information.<sup>2</sup> Patients also report that reading notes through the portal provided a sense of control of one's health,<sup>3-6</sup> improved adherence to treatment and follow-up plans,<sup>3</sup> and bolstered understanding of their disease and treatment.<sup>4,5</sup> Across studies, only 3%–16% of adult patients reported increased worry or confusion after reading their clinical notes.<sup>3,4,6,7</sup>

Past studies have found socioeconomic disparities in which patients access portals. Patients and parents with lower education, historically minoritized status, and lower income are less likely to use portals, <sup>8–11</sup> despite 85% of US adults owning cell phones capable of accessing the portal<sup>12</sup> and 93% having internet access.<sup>13</sup> However, prior studies have not evaluated the role of clinicians in encouraging portal use to patients. In other areas of communication research, clinician endorsement of interventions is critical to support patient engagement. For example, active clinician endorsement of question prompt lists dramatically increases the frequency of questions asked by patients.<sup>14</sup>

In this study, we aimed to identify factors associated with online electronic medical record (EMR) access in a nationally representative sample of the 2020 Health Information National Trends Survey 5 (HINTS 5) cycle 4. We hypothesized that clinician encouragement might influence online EMR use behaviors of patients.

#### OBJECTIVE

The aim of this study was to understand the influence of clinician encouragement and sociodemographic factors on whether patients access their EMR online.

#### MATERIALS AND METHODS

#### Methods

We report this study following Strengthening the Reporting of Observational Studies in Epidemiology guidelines.<sup>15</sup>

#### Data sources

We used data from HINTS, a nationally representative crosssectional survey of noninstitutionalized adults in the United States aged 18 and over. The HINTS survey was developed in 2003 by the Health Communication and Informatics Research Branch of the Division of Cancer Control and Population Sciences. The HINTS survey collects information about respondents' attitudes about health, cancer, health information, and technology. For additional details about HINTS methodology visit hints.cancer.gov.

#### Setting

We used data from HINTS 5, cycle 4 which was collected between February and June 2020 using a single-mode mail survey. The survey was sent to 15 350 randomly selected addresses in the United States, with over-sampling in census tracts with a higher proportion (34% or higher) of African American and Hispanic residents. Addresses were selected irrespective of medical care history or current care involvement.

#### Study size

After adjusting for unresolved households (ie, the households that never return a survey or refuse, or have mailings returned because they were undeliverable), the response rate was 36.7%, with a higher response rate among lower historically minoritized community (40.3%) versus higher-historically minoritized community (27.2%). Of 3777 respondents, we removed 498 responses that lacked complete data on all covariates, leaving a sample size of 3279 respondents in our final analytic dataset. Compared to the study sample, the excluded respondents were more likely to be over 70 years old, female, Hispanic or non-Hispanic Black, not college graduates, have lower incomes, have a history of cancer, and reside in rural locations (Supplementary Appendix S1).

#### Bias

The HINTS survey oversampled addresses in areas with higher densities of African Americans and Hispanic Americans to attempt to overcome potential nonresponse bias in these populations. In addition, to examine and adjust for potential bias in our study design, we examined differences in characteristics between respondents in our final analytic dataset and those that were removed due to missing data (Supplementary Appendix S1). Finally, we employed survey weights in analyses to ensure national representativeness of study results. Analyses were conducted in STATA 17.0 using sample weights to produce population point estimates and a set of 50 jackknife replicate weights to compute variance estimates. Weights were supplied by the Westat for the HINTS survey to account for nonresponse and noncoverage biases to the fullest extent possible. Weights were calculated based on probability of being sampled within the household, household nonresponse, and demographic information from the 2018 American Community Survey.

#### Variables

Our main outcome variables in this study were (1) a binary variable capturing whether respondents ever accessed their online EMR (ie, 1 if the answer to question "How many times did you access your online medical record in the last 12 months?" was greater than 0, 0 if the answer was 0), and (2) whether respondents were encouraged to use their portal (ie, 1 if the answer to question "Have any of your health care providers, including doctors, nurses, or office staff, ever encouraged you to use an online medical record?" was yes, 0 if the answer was no).

Our main independent variables of interest were the following self-reported respondent characteristics: age, gender assigned at birth, race and ethnicity, education, history of cancer, and history of chronic disease. For age, we created an ordinal variable with the following age groups: 18–29, 30– 39, 40–49, 50–59, 60–69, and over 70. For sex, we created a binary variable indicating whether the respondent was listed as male or female on their original birth certificate. For race and ethnicity, we created 5 binary variables following the 5 US census categories for race and ethnicity based on selfreport: Hispanic, non-Hispanic Asian, non-Hispanic Black or African American, non-Hispanic White, and non-Hispanic other. For education, we created a binary variable which indicated whether the respondent completed college or postgraduate education. For history of cancer, we created a binary variable indicating whether the respondent reported ever being diagnosed with cancer. For history of chronic disease, we created a binary variable indicating whether the respondent indicated they had ever had diabetes, high blood pressure, heart condition, lung disease, or depression. We chose to operationalize chronic disease as a binary variable due to high collinearity among disease prevalence. We present results with chronic diseases examined separately in Supplementary Appendix S2.

Respondents who reported not accessing their EMR in the past 12 months were asked whether they did not access their EMR because they: (1) preferred to speak to their health care provider directly, (2) did not have a way to access the website, (3) did not have a need to use their EMR, (4) were concerned about the privacy or security of the website that hosted their EMR, (5) did not have an online medical record, (6) found it difficult to log in (eg, having trouble remembering a password), (7) were not comfortable or experienced with computers, and (7) had more than 1 online medical record. We created binary variables yes/no for each response and examined bivariate distributions of responses by respondent characteristics.

#### Statistical methods

We ran descriptive analyses to understand the distribution of our main outcome variables by respondent characteristics. We performed unadjusted bivariate logistic regression to assess the relationship between our main outcome variables and respondent characteristics. Then, we ran 3 sets of multivariate logistic regression models. The first model regressed respondent characteristics on EMR use. The second model regressed respondent characteristics and encouragement on EMR use. The third model regressed respondent characteristics on encouragement. Finally, of respondents who did not access their EMRs online, we used 2-tailed chi-squared tests to determine statistical significance of differences in reported reasons for nonuse by groups.

#### RESULTS

The estimated percentage of US adults who accessed their EMR was 41% and the estimated percentage of US adults who were encouraged to access their EMR was 51%. EMR users were more likely to be female compared to male, non-Hispanic White compared to racial and ethnic historically minoritized Americans, have a college degree or higher, have higher incomes, and have a history of cancer (Table 1) Americans who reported being encouraged to access their EMR online were more likely to be female compared to male, non-Hispanic White compared to racial and ethnic historically minoritized Americans, have a college degree or higher, have higher incomes, and have a history of cancer (Table 1).

In bivariate logistical regression, we observed significant differences in the rate of both access and encouragement by age, gender, ethnicity, race, education, income, and history of cancer (Table 2). Participants were more likely to have used the EMR if they received encouragement from their clinical team (odds ratio [OR], 11.5; 95% confidence interval [CI], 8.5–15.3), were female compared to male (OR, 1.7; 95% CI,

1.4–2.1), were non-Hispanic White compared to racial and ethnic historically minoritized Americans (OR, 1.8; 95% CI, 1.3–2.3), had a college degree or higher (OR, 2.5; 95% CI, 2.0–3.2), had higher income, and had a history of cancer (OR, 1.7; 95% CI, 1.2–2.4). Similarly, reporting clinician encouragement to use EMR was associated with being female compared to male (OR, 1.6; 95% CI, 1.2–2.0), non-Hispanic White compared to racial and ethnic historically minoritized Americans (OR, 1.3; 95% CI, 1.0–1.7), having college education or higher (OR, 1.9; 95% CI, 1.5–2.4), having higher-income categories, and having a history of cancer (OR, 1.6; 95% CI, 1.2–2.3) (Table 2).

In multivariate analyses we found that, controlling for other patient characteristics, EMR use was associated with female gender (OR, 1.9; 95% CI, 1.5–2.4), having a college degree or higher (OR, 2.0; 95% CI, 1.5–2.7), having higherincome categories, and having a history of cancer (OR, 1.9, 95% CI, 1.3–2.6; Table 3, model 1). Hispanic ethnicity was inversely associated with EMR use, compared to non-Hispanic White (OR, 0.5; 95% CI, 0.3–0.7). When including encouragement in the model, we found that income was no longer a significant covariate (Table 3, model 2). Controlling for all other factors, receiving encouragement was associated with female gender (OR, 1.7; 95% CI, 1.3–2.3), having a college degree or higher (OR, 1.5; 95% CI, 1.1–2.0), having annual incomes greater than 20k, and having a history of cancer (OR, 1.8; 95% CI, 1.3–2.5; Table 3, model 3).

The most frequently cited reasons for not accessing the EMR online were: preferring to speak directly with their provider (68%), followed by no need (61%), no online medical record (32%), privacy/security concerns (24%), uncomfortable or inexperienced with computers (19%), no way to access (20%), log-in problems (18%), and multiple medical records (5%). In Table 4, we present these reasons for not accessing the EMR, stratified by participant characteristics.

#### DISCUSSION

In this nationally representative dataset, we found overall low rates of online EMR use with statistically significant disparities by gender, race and ethnicity, and education, which may be partially driven by differential rates of encouragement from clinicians. We found that patients who were male, Hispanic, had less education, or had incomes under \$50 000 were significantly less likely to access EMR than their female, White, college-educated, and higher-income counterparts. These findings are consistent with prior studies, which showed that education, gender, and race and ethnicity were associated with accessing portals<sup>16,17</sup> and e-communicating with clinicians in both pediatric and adult medicine.<sup>8-11,18</sup> These disparities matter because portal use is associated with improved communication, patient understanding, and quality of care.<sup>10,19</sup> Furthermore, in 2013, the Meaningful Use program was introduced to ensure the widespread availability of portals in the US healthcare system. However, our findings suggest that nearly 10 years later, portal use overall is still relatively low, especially among marginalized populations.

Unlike these prior studies, our analysis shows that clinician encouragement plays an important role in encouraging patients to access their EMR through online patient portals. Our analyses suggest that, even after controlling for structural inequities that could contribute to low portal use among marginalized groups, encouragement still plays an outsized

		Accessed		Encouraged		
	Total population percent	Percent of subgroup	P-value	Percent of subgroup	P-value	
Total		41		51		
Age group			.053		.101	
18–29	19	36		44		
30–39	16	42		54		
40–49	19	46		52		
50-59	19	48		58		
60–69	15	40		50		
Over 70	12	34		47		
Sex assigned at birth			<.001		.001	
Male	49	35		45		
Female	51	48		56		
Race/ethnicity			<.001		.021	
Hispanic	17	24		41		
Non-Hispanic Asian	5.2	46		52		
Non-Hispanic Black or African American	11	39		48		
Non-Hispanic White	64	46		54		
Non-Hispanic other	3.2	38		66		
Education			<.001		<.001	
Less than college	69	34		46		
College degree or higher	31	57		62		
Income			<.001		<.001	
<20k	14	28		33		
20–49k	24	34		45		
50–99k	31	43		52		
Over 100k	31	52		62		
Cancer history			.002		.007	
History	8.9	53		62		
No history	91	40		50		
Chronic disease			<.001		.001	
History	58	48		55		
No history	42	32		45		
Urban/rural		-	.351	-	.672	
Urban	88	42		51		
Rural	12	37		50		

Table 1. Patient characteristics by patient online EMR	R use and encouragement, population estimates (n=3279)
--	--

Note: P-values indicate statistical significance of Pearson's chi-squared tests for difference between accessed/encouraged by subgroup membership.

role in predicting online EMR use. This is a critical finding since clinician behavior is a modifiable factor with significant room for improvement—only 51% of all respondents reported being encouraged by their clinician to use the EMR. Prior studies of health behavior change suggest that clinicians can use 5 key strategies to encourage use of portals: "emphasizing patient ownership; partnering with patients; identifying small steps; scheduling frequent follow-up visits to cheer successes, problem solve, or both; and showing caring and concern for patients."<sup>20</sup> Future studies must identify strategies to successfully encourage patients in using portals for accessing their EHI. Furthermore, healthcare organizations must support clinicians in this endeavor by developing standardized workflows and accessible technologies to support portal use.

Respondents who were Hispanic, male, and noncollege educated were less likely to access their online EMR, even after controlling for encouragement. These disparities in access could be related to patient preferences, structural barriers such as internet and computer access, technological literacy, and characteristics of the portal interface, such as language, user interface, and complexity of content. Our findings also suggest that these disparities could partially be attributed to clinicians' actions, influenced by implicit biases that influence the tone, content, or frequency of discussions encouraging online EMR use.

Although Hispanic Americans were least likely to access the online EMR (24% compared to 46% of non-Hispanic White Americans) compared to other racial and ethnic groups, they were not less likely to be encouraged to use the portal in multivariable analysis. Prior work suggests that language barriers could play a role in portal access. Hispanic patients demonstrated lower internet health informationseeking behaviors, associated partially with foreign-birth and language preferences.<sup>21,22</sup> While some components of portals can be translated to other languages, most of the content is in English. Taken together with our finding that male, noncollege-educated, and lower-income respondents are less likely to be encouraged to access the EMR online, our study suggests that Hispanic populations are least likely to access EMR due to multiple confounding factors, exacerbated even more by the fact that Hispanic men are less likely than other men to enroll and complete college<sup>23</sup> and undocumented Hispanic immigrants are less likely to access any healthcare services<sup>24</sup> compared to other racial and ethnic groups. The role of multiple confounding factors is supported by our findings that Hispanic ethnicity is associated with encouragement in bivariate analysis, but not when adjusted for these other

Table 2.	Unadjusted	bivariate	logistic	regression	results	(OR	[95%	CI])
----------	------------	-----------	----------	------------	---------	-----	------	------

	Accessed	Encouraged
Encouraged	11.45*** [8.55, 15.34]	
Age group (ref: not in the category)		
18–29	0.75 [0.51, 1.11]	0.72 [0.49, 1.06]
30-39	1.00 [0.71, 1.41]	1.16 [0.83, 1.63]
40–49	1.24 [0.90, 1.70]	1.04 [0.74, 1.45]
50-59	1.41* [1.03, 1.93]	1.39* [1.05, 1.86]
60–69	0.92 [0.69, 1.22]	0.94 [0.73, 1.23]
Over 70	0.72* [0.56, 0.93]	0.83 [0.64, 1.07]
Gender assigned at birth (ref: male)		
Female	1.69*** [1.37, 2.13]	1.56** [1.20, 2.00]
Race/ethnicity (ref: not in the category)		
Hispanic	0.39*** [0.28, 0.55]	$0.60^*$ [0.41, 0.88]
Non-Hispanic Asian	1.21 [0.64, 2.30]	1.02 [0.56, 1.87]
Non-Hispanic Black or African American	0.88 [0.59, 1.29]	0.87 [0.60, 1.25]
Non-Hispanic White	1.75*** [1.33, 2.32]	1.32* [1.04, 1.68]
Non-Hispanic other	0.86 [0.42, 1.79]	1.89 [0.89, 4.03]
Education (ref: less than college)		
College degree or higher	2.53*** [2.00, 3.20]	1.93*** [1.53, 2.43]
Income (ref: not in the category)		
<20k	0.51** [0.33, 0.80]	0.41*** [0.29, 0.59]
20–49k	0.65* [0.47, 0.90]	0.75* [0.56, 0.99]
50–99k	1.06 [0.85, 1.32]	1.07 [0.81, 1.41]
Over 100k	1.90*** [1.41, 2.55]	1.95*** [1.45, 2.61]
Cancer history (ref: no history)		
History of cancer	1.69** [1.22, 2.35]	1.63** [1.15, 2.32]
Chronic disease history (ref: no history)		
History of chronic disease	1.95*** [1.49, 2.56]	1.46** [1.17, 1.83]
Urban/rural (ref: urban)		
Rural	0.82 [0.54, 1.25]	0.94 [0.69, 1.27]

*Note: P*-values indicate statistical significance for difference between accessed/encouraged by subgroup membership; n = 3279. \* P < .05. \*\* P < .01. \*\*\* P < .001.

Table 3. Multivariate regression model results (adjusted odds ratios [95% CI])

	Encouraged	Accessed (not incl. encouragement)	Accessed (including encouragement)		
Encouraged	N/A	N/A	10.34*** [7.66, 13.97]		
Age group (ref: 18–29)					
30-39	1.20 [0.75, 1.92]	0.85 [0.50, 1.44]	0.74 [0.40, 1.37]		
40-49	1.09 [0.67, 1.77]	1.06 [0.67, 1.68]	1.06 [0.60, 1.87]		
50-59	1.26 [0.76, 2.07]	1.01 [0.65, 1.57]	0.88 [0.55, 1.41]		
60–69	1.01 [0.67, 1.53]	0.73 [0.45, 1.17]	0.68 [0.39, 1.17]		
Over 70	0.88 [0.57, 1.33]	0.55** [0.35, 0.85]	0.52* [0.30, 0.89]		
Gender assigned at Birth (ref: male)					
Female	1.74*** [1.31, 2.32]	1.88*** [1.47, 2.41]	1.58** [1.20, 2.07]		
Race/ethnicity (ref: non-Hispanic White)					
Hispanic	0.8 [0.53, 1.20]	0.49*** [0.33, 0.71]	0.46** [0.28, 0.76]		
Non-Hispanic Asian	0.87 [0.48, 1.59]	0.86 [0.43, 1.71]	0.87 [0.45, 1.69]		
Non-Hispanic Black or African American	0.95 [0.65, 1.40]	0.79 [0.53, 1.19]	0.78 [0.46, 1.32]		
Non-Hispanic Other	2.13 [0.86, 5.29]	0.78 [0.32, 1.89]	0.49 [0.17, 1.37]		
education (ref: less than college)					
College degree or higher	1.45* [1.07, 1.96]	1.98*** [1.47, 2.65]	1.92*** [1.36, 2.69]		
Income (ref: <20k)					
20–49k	1.79** [1.19, 2.71]	1.31 [0.79, 2.18]	0.99 [0.52, 1.87]		
50–99k	2.36*** [1.55, 3.58]	1.84** [1.18, 2.86]	1.26 [0.76, 2.08]		
Over 100k	3.64*** [2.20, 6.04]	2.60*** [1.56, 4.34]	1.57 [0.88, 2.79]		
Cancer history (ref: no history)					
History of cancer	1.77** [1.26, 2.49]	1.85*** [1.32, 2.59]	1.54* [1.04, 2.29]		
Chronic disease (ref: no history)					
History of chronic disease	1.65*** [1.30, 2.10]	2.38*** [1.78, 3.19]	2.28*** [1.65, 3.15]		
Urban/rural (ref: urban)					
Rural	0.96 [0.70, 1.32]	0.75 [0.51, 1.10]	0.7 [0.43, 1.13]		

Notes: weighted estimates; n = 3279. \* P < .05. \*\* P < .01. \*\*\* P < 0.001.

Table 4. Percentage of respondents who did not access their EMR online and reasons for not accessing (population point estimates)

	Prefer to speak directly (%)	No need (%)	No record (%)	Privacy/security concern (%)	Comfort (%)	No way to access website (%)	Log-in problems (%)	Multiple records (%)
All respondents	68	61	32	24	19	20	18	5.2
Age group ( <i>P</i> -value <sup>a</sup> )	.006	.244	.549	.329	<.001	.183	.303	.042
18–29	58	66	38	17	7.8	22	12	1.9
30–39	61	61	31	22	10	14	18	5.9
40-49	58	61	32	26	14	16	21	6.4
50-59	76	67	31	26	18	19	17	4.8
60–69	81	53	31	33	39	24	26	9.6
Over 70	82	54	27	26	44	31	20	3.6
Biological gender (P-value)	.548	.030	.784	.036	.521	.134	.009	.469
Male	67	65	33	21	19	23	15	4.7
Female	69	56	32	28	21	18	22	5.9
Race/ethnicity (P-value)	.445	<.001	.228	.464	.832	.205	.608	.002
Hispanic	63	55	28	23	17	20	21	3.2
Non-Hispanic Asian	69	50	34	23	21	15	21	4.4
Non-Hispanic Black or African American	76	40	24	30	20	33	20	4.3
Non-Hispanic White	69	69	36	24	20	19	17	5.3
Non-Hispanic other	61	56	28	12	13	20	14	22
Education ( <i>P</i> -value)	.014	.854	.598	.733	<.001	.190	.162	.234
Less than college	71	61	32	25	23	22	19	4.8
College degree or higher	59	61	34	23	6.9	17	16	6.6
Income ( <i>P</i> -value)	.002	.014	.906	.464	<.001	.136	.203	.974
<20k	75	48	30	29	26	27	24	4.8
20–49k	78	62	31	27	28	24	20	5.5
50–99k	64	60	34	23	18	19	18	4.9
Over 100k	58	70	33	21	9.3	15	14	5.5
Cancer ( <i>P</i> -value)	.002	.429	.065	.623	<.001	.057	.007	.768
No history	67	62	33	24	18	20	17	5.3
History of cancer	82	56	23	26	37	29	31	4.7
Chronic disease (P-value)	.060	.644	.5.50	.015	.040	.640	.092	.469
No history	63	62	34	20	1.5	19	1.5	4.7
History of chronic disease	72	60	31	28	23	21	21	5.7
Urban/rural ( <i>P</i> -value)	.960	.8.57	.747	.031	.097	.944	.023	.123
Urban	68	61	32	25	19	20	19	5.6
Rural	68	62	34	17	26	21	12	2.5
N	1651	1626	1598	1659	1628	1640	1607	1595

Notes: Numbers represent the percent of each subpopulation who indicated that this was a reason for not accessing their EMR online, reasons for nonuse limited to subsample of respondents who reported never accessing their EMR online; subsample sizes differ by reasons provided due to missing data, totals may not sum to 100% due to rounding.

a' P-values indicate statistical significance of chi-squared test of difference between group proportions; bolded P-values highlight P-values <.05.

factors. As such, encouragement alone is unlikely to mitigate these multiple barriers. Future research should strive to more fully understand the barriers and facilitators to EMR use among Hispanic patients.

In addition to these sociodemographic factors, accessing online EMR and receiving encouragement to access EMR were associated with the history of both cancer and chronic disease. For patients with ongoing chronic health issues, this access could help them to better understand their treatment plans, prevention strategies, and support a sense of control over their health.<sup>2-6</sup> Patients with a history of cancer will have ongoing health complications and late effects due to the cancer and its treatments.<sup>25,26</sup> Many cancer survivors do not receive adequate monitoring of these complications, and many survivors struggle to access survivorship clinics with expertise in the management of these long-term health issues.<sup>27–29</sup> By leveraging the EMR in conjunction with other applications, clinicians might improve engagement and selfmanagement for these patients with chronic, ongoing health needs. However, current work has mostly focused on encouraging enrollment and access to these online portals. Future work should strive to understand how to make these online portals most useful for patients with history of chronic disease and/or cancer, rather than focusing on access alone.

Notably, data collection was conducted at the beginning of the COVID-19 pandemic. The use of telemedicine, remote patient monitoring, virtual triage and screening tools, and portals to access EHI came to the forefront at the start of the surge.<sup>30-33</sup> For example, when COVID testing became available, due to the volume of testing, it was not feasible for many practices to call patients with results, and some offices mandated portal access to get results. However, recent data also indicate that low-income families had limited internet access, connectivity issues, lacked internet-connecting devices, and faced financial hardships due to the pandemic. Black, Hispanic, and low-income families were hit hardest by digital inequity.<sup>34</sup> These factors compounded with the lack of digital literacy previously documented among Hispanic populations in the United States, could have potentially limited online access to EMR. However, the true impact of COVID on these results remains unknown and could have affected different populations uniquely. Future research is needed.

This study should be interpreted in light of limitations. First, this questionnaire did not assess for multiple structural barriers to online EMR use. Behavior is influenced by multilevel factors, and future studies should examine the role of these additional factors. Second, the constructs of "race" and "ethnicity" are incomplete and ever-changing social constructs that are used to label diverse populations of people with complex, interacting identities, societal structures, cultural norms, and heritages. For example, the category of "Hispanic" conflates many different people groups. Prior studies have found differences in health information-seeking behavior among people from Hispanic and Latin cultures, as well as differences between US-born versus foreign-born patients.<sup>35,36</sup> Future studies should identify factors contributing to this disparity in portal access among Hispanic and Latin subpopulations. This same recommendation must be applied to the racial categories as those do not take into consideration the complexity of different racial groups that represent the diaspora of Black/African American, Asian, Pacific Islander, and Indigenous communities. Despite this limitation, we believe that the use of racial and ethnic categories can highlight areas where disparities exist and serve as proxy for other factors that are difficult to measure, such as structural or interpersonal racism. Third, the HINTS data are crosssectional and only represent 1 point in time. Thus, the associations found in this article are not necessarily causal. Likewise, data collection was conducted at the beginning of the COVID-19 pandemic. This may have affected access patterns both in favor of, and against both access and survey completion. Finally, while the response rate of 36.7% could lead to selection bias, we believe that the sampling and weighting strategy employed by HINTS administrators minimized this bias and improved representativeness and generalizability. Future studies should aim to confirm these results with more detail and higher response rates in local settings. Lastly, while not a limitation, these results precede the enactment of the 21st Century Cures Act mandate in April 2021 which required that all electronic portals be available to patients in a timely manner and free of charge. While many organizations are using portals to satisfy this requirement, we cannot observe for changes related to this mandate.

#### CONCLUSION

Clinician encouragement of EMR use is strongly associated with patients accessing their EMR online. In addition to finding persistent disparities in online EMR use related to gender, race and ethnicity, education, and income found in previous HINTS cycles, we also found disparities in clinician encouragement of portal use among similar subgroups. Future studies should evaluate the role of clinician behaviors in facilitating portal use, and how to facilitate these supportive clinician behaviors. The 21st Century Cures Act guarantees the widespread access to EHI; therefore, it is imperative to ensure that the benefits of the policy change are accessible to all patient populations.

#### FUNDING

This research received no specific grant from any funding agency in the public, commercial or not-for-profit sectors.

#### AUTHOR CONTRIBUTIONS

BAS and SL contributed to the conception and design of the work, acquisition of data, interpreting data, and drafting and revising the manuscript. JEBB, AES, and JWM contributed to the conception and design of the work, interpretation of data, and revision of the manuscript. All authors approved the final manuscript and agree to be accountable for all aspects of the work.

#### SUPPLEMENTARY MATERIAL

Supplementary material is available at JAMIA Open online.

#### **CONFLICT OF INTEREST STATEMENT**

None declared.

#### DATA AVAILABILITY

This study analyzed publicly available data from the HINTS dataset. This data can be accessed from https://hints.cancer.gov/data/download-data.aspx.

#### REFERENCES

- 1. 21st Century Cures Act: Interoperability, Information Blocking, and the ONC Health IT Certification Program. 45 CFR Parts 170 and 171. Issuing Organization: Department of Health and Human Services; 2020.
- Santos AD, Caine V, Robson PJ, Watson L, Easaw JC, Petrovskaya
  O. Oncology patients' experiences with novel electronic patient
  portals to support care and treatment: qualitative study with early
  users and nonusers of portals in Alberta, Canada. *JMIR Cancer*2021; 7 (4): e32609.
- 3. Delbanco T, Walker J, Bell SK, *et al.* Inviting patients to read their doctors' notes: a quasi-experimental study and a look ahead. *Ann Intern Med* 2012; 157 (7): 461–70.
- Weinert C. Giving doctors' daily progress notes to hospitalized patients and families to improve patient experience. *Am J Med Qual* 2017; 32 (1): 58–65. [published Online First: 2015/10/11].
- DesRoches CM, Bell SK, Dong Z, et al. Patients managing medications and reading their visit notes: a survey of OpenNotes participants. Ann Intern Med 2019; 171 (1): 69–71.
- Mishra VK, Hoyt RE, Wolver SE, Yoshihashi A, Banas C. Qualitative and quantitative analysis of patients' perceptions of the patient portal experience with OpenNotes. *Appl Clin Inform* 2019; 10 (1): 10–8.
- Walker J, Leveille S, Bell S, *et al.* OpenNotes after 7 years: patient experiences with ongoing access to their clinicians' outpatient visit notes. *J Med Internet Res* 2019; 21 (5): e13876.
- Yang R, Zeng K, Jiang Y. Prevalence, factors, and association of electronic communication use with patient-perceived quality of care from the 2019 health information national trends survey 5cycle 3: exploratory study. *J Med Internet Res* 2022; 24 (2): e27167.
- 9. LeLaurin JH, Nguyen OT, Thompson LA, *et al.* Disparities in pediatric patient portal activation and feature use. *JAMIA Open* 2021; 4 (3): 00ab086.
- Schultz CL, McCahan SM, Lewis AM, Bunnell HT, Alderfer MA. Online patient portal use by caregivers in pediatric oncology: are we widening sociodemographic disparities? *Pediatr Blood Cancer* 2021; 68 (12): e29373.
- Parker R, Goldbloom EB, Mitsakakis N, Terekhov I, Zuijdwijk C. Social determinants of health associated with patient portal use in pediatric diabetes. *Can J Diabetes* 2022; 46 (6): 586–93.

- Pew Research Center. Secondary Pew Research Center—Mobile Fact Sheet; 2021. https://www.pewresearch.org/internet/fact-sheet/ mobile/.
- Pew Research Center. Secondary Pew Research Center—Internet/ Broadband Fact Sheet; 2021. https://www.pewresearch.org/internet/fact-sheet/internet-broadband/.
- 14. Clayton JM, Butow PN, Tattersall MH, *et al.* Randomized controlled trial of a prompt list to help advanced cancer patients and their caregivers to ask questions about prognosis and end-of-life care. *J Clin Oncol* 2007; 25 (6): 715–23.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP; STROBE Initiative. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. *Ann Intern Med* 2007; 147 (8): 573–7.
- Elkefi S, Yu Z, Asan O. Online medical record nonuse among patients: data analysis study of the 2019 health information national trends survey. J Med Internet Res 2021; 23 (2): e24767.
- 17. Anthony DL, CAmpos-Castillo C, Lim PS. Who isn't using patient portals and why? Evidence and implications from a national sample of US adults. *Health Aff (Millwood)* 2018; 37 (12): 1948–54.
- Lin SC, Lyles CR, Sarkar U, Adler-Milstein J. Are patients electronically accessing their medical records? Evidence from national hospital data. *Health Aff (Millwood)* 2019; 38 (11): 1850–7.
- Acholonu RG, Raphael JL. The influence of the electronic health record on achieving equity and eliminating health disparities for children. *Pediatr Ann* 2022; 51 (3): e112–e117.
- Greene J, Hibbard JH, Alvarez C, Overton V. Supporting patient behavior change: approaches used by primary care clinicians whose patients have an increase in activation levels. *Ann Fam Med* 2016; 14 (2): 148–54.
- 21. Peña-Purcell N. Hispanics' use of Internet health information: an exploratory study. *J Med Libr Assoc* 2008; 96 (2): 101–7.
- Massey PM, Langellier BA, Sentell T, Manganello J. Nativity and language preference as drivers of health information seeking: examining differences and trends from a U.S. population-based survey. *Ethn Health* 2017; 22 (6): 596–609.
- 23. Saenz VB, Ponjuan L. The vanishing Latino male in higher education. J Hispanic High Educ 2009; 8 (1): 54–89.
- 24. Ortega AN, Fang H, Perez VH, *et al.* Health care access, use of services, and experiences among undocumented Mexicans and other Latinos. *Arch Intern Med* 2007; 167 (21): 2354–60.
- 25. Nekhlyudov L, Mollica MA, Jacobsen PB, Mayer DK, Shulman LN, Geiger AM. Developing a quality of cancer survivorship care

framework: implications for clinical care, research, and policy. J Natl Cancer Inst 2019; 111 (11): 1120–30.

- Benci JL, Vachani CC, Hampshire MK, *et al.* Factors influencing delivery of cancer survivorship care plans: a national patterns of care study. *Front Oncol* 2019; 9: 1577.
- Zheng DJ, Sint K, Mitchell HR, Kadan-Lottick NS. Patterns and predictors of survivorship clinic attendance in a population-based sample of pediatric and young adult childhood cancer survivors. J Cancer Surviv 2016; 10 (3): 505–13.
- Lindell RB, Koh SJ, Alvarez JM, et al. Knowledge of diagnosis, treatment history, and risk of late effects among childhood cancer survivors and parents: The impact of a survivorship clinic. Pediatr Blood Cancer 2015; 62 (8): 1444–51.
- Ford JS, Chou JF, Sklar CA. Attendance at a survivorship clinic: impact on knowledge and psychosocial adjustment. J Cancer Surviv 2013; 7 (4): 535–43.
- Portz JD, Brungardt A, Shanbhag P, *et al.* Advance care planning among users of a patient portal during the COVID-19 pandemic: retrospective observational study. *J Med Internet Res* 2020; 22 (8): e21385.
- Nath B, Williams B, Jeffery MM, et al. Trends in electronic health record inbox messaging during the COVID-19 pandemic in an ambulatory practice network in New England. JAMA Netw Open 2021; 4 (10): e2131490.
- 32. Mahajan V, Singh T, Azad C. Using telemedicine during the COVID-19 pandemic. *Indian Pediatr* 2020; 57 (7): 658–61.
- Cassar MR, Borg D, Camilleri L, et al. A novel use of telemedicine during the COVID-19 pandemic. Int J Infect Dis 2021; 103: 182–7.
- 34. Katz V, Rideout V. Learning at Home While Under-Connected: Lower-Income Families During the COVID-19 Pandemic. Secondary Learning at Home While Under-Connected: Lower-Income Families During the COVID-19 Pandemic June 24, 2021. https:// www.newamerica.org/education-policy/reports/learning-at-homewhile-underconnected/.
- 35. Gonzalez M, Sanders-Jackson A, Wright T. Web-based health information technology: access among latinos varies by subgroup affiliation. *J Med Internet Res* 2019; 21 (4): e10389.
- 36. Gonzalez M, Sanders-Jackson A, Emory J. Online health information-seeking behavior and confidence in filling out online forms among latinos: a cross-sectional analysis of the california health interview survey, 2011–2012. J Med Internet Res 2016; 18 (7): e184.